

JGR Atmospheres

RESEARCH ARTICLE

10.1029/2019JD030581

Key Points:

- Multimodel mean efficacies computed using effective radiative forcing are close to one for major anthropogenic drivers, but there is notable intermodel spread
- Surface temperature-driven
 radiative feedbacks are generally not
 constant through time across forcing
 experiments
- PDRMIP results suggest that the efficacy impact on equilibrium climate sensitivity derived from the historical period is limited to ±0.4 °C or better

Supporting Information:

Supporting Information S1

Correspondence to: P. M. Forster.

p.m.forster@leeds.ac.uk

Citation:

Richardson, T. B., Forster, P. M., Smith, C. J., Maycock, A. C., Wood, T., Andrews, T., et al. (2019). Efficacy of climate forcings in PDRMIP models. *Journal of Geophysical Research: Atmospheres*, *124*, 12,824–12,844. https://doi.org/10.1029/2019JD030581

Received 3 MAR 2019 Accepted 16 NOV 2019 Accepted article online 20 NOV 2019 Published online 11 DEC 2019

©2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Efficacy of Climate Forcings in PDRMIP Models

- T. B. Richardson¹, P. M. Forster¹, C. J. Smith¹, A. C. Maycock¹, T. Wood¹,
- T. Andrews², O. Boucher³, G. Faluvegi⁴, D. Fläschner⁵, Ø. Hodnebrog⁶, M. Kasoar⁷,
- A. Kirkevåg⁸ D, J.-F. Lamarque⁹ D, J. Mülmenstädt¹⁰ D, G. Myhre⁶ D, D. Olivié⁸,
- R. W. Portmann¹¹, B. H. Samset⁶, D. Shawki⁷, D. Shindell¹², P. Stier¹³,
- T. Takemura¹⁴, A. Voulgarakis⁷, and D. Watson-Parris¹³

¹Priestley International Centre for Climate, University of Leeds, Leeds, UK, ²Met Office Hadley Centre, Exeter, UK, ³Institut Pierre-Simon Laplace, CNRS/Sorbonne Université, Paris, France, ⁴NASA Goddard Institute for Space Studies and Center for Climate Systems Research, Columbia University, New York, NY, USA, ⁵Atmosphere in the Earth System, Max-Planck-Institut für Meteorologie, Hamburg, Germany, ⁶CICERO Center for International Climate and Environmental Research, Oslo, Norway, ⁷Department of Physics, Imperial College London, London, UK, ⁸Research and Development Department, Norwegian Meteorological Institute, Oslo, Norway, ⁹NCAR/UCAR, Boulder, CO, USA, ¹⁰Clouds and Global Climate, Universität Leipzig, Leipzig, Germany, ¹¹Earth System Research Laboratory, National Oceanic and Atmospheric Administration, Boulder, CO, USA, ¹²Earth & Ocean Sciences, Duke University, Durham, NC, USA, ¹³Atmospheric, Oceanic and Planetary Physics, Department of Physics, University of Oxford, Oxford, UK, ¹⁴Center for Oceanic and Atmospheric Research, Kyushu University, Fukuoka, Japan

Abstract Quantifying the efficacy of different climate forcings is important for understanding the real-world climate sensitivity. This study presents a systematic multimodel analysis of different climate driver efficacies using simulations from the Precipitation Driver and Response Model Intercomparison Project (PDRMIP). Efficacies calculated from instantaneous radiative forcing deviate considerably from unity across forcing agents and models. Effective radiative forcing (ERF) is a better predictor of global mean near-surface air temperature (GSAT) change. Efficacies are closest to one when ERF is computed using fixed sea surface temperature experiments and adjusted for land surface temperature changes using radiative kernels. Multimodel mean efficacies based on ERF are close to one for global perturbations of methane, sulfate, black carbon, and insolation, but there is notable intermodel spread. We do not find robust evidence that the geographic location of sulfate aerosol affects its efficacy. GSAT is found to respond more slowly to aerosol forcing than CO_2 in the early stages of simulations. Despite these differences, we find that there is no evidence for an efficacy effect on historical GSAT trend estimates based on simulations with an impulse response model, nor on the resulting estimates of climate sensitivity derived from the historical period. However, the considerable intermodel spread in the computed efficacies means that we cannot rule out an efficacy-induced bias of ± 0.4 K in equilibrium climate sensitivity to CO₂ doubling when estimated using the historical GSAT trend.

Plain Language Summary Does the climate respond in the same way to carbon dioxide as it does to methane or aerosol changes? The simple way of thinking about forcing and response in the Earth system assumes that it does, such that, a Watt per square meter forcing from CO_2 has the same response as an equivalent forcing from aerosols. Recent work has suggested that this might not be true and that differences in how effective different forcings are at increasing surface temperature (their efficacy) may account for a low estimate of climate sensitivity when examining historical change. We show this all depends on how you estimate your Watts per meter squared forcing in the first place. Using the effective radiative forcing concept to estimate forcing strength makes temperature changes far more predictable, and a lot of these issues with efficacy variation are not as pronounced as they were with earlier definitions.

1. Introduction

Quantifying the effectiveness of different physical drivers in altering surface temperature is important for understanding historical and future temperature trends (Kummer & Dessler, 2014; Marvel et al., 2015). Global mean near-surface air temperature (GSAT) change is typically understood in terms of the well-established radiative forcing and response framework (equation (1); Hansen et al., 1997; Boer & Yu, 2003; Gregory, 2004; Myhre et al., 2013). Radiative forcing represents the change in the Earth's energy balance due to an imposed perturbation, such as increasing greenhouse gases. In response to an imposed forcing

which is held constant, GSAT will increase or decrease until the climate system reaches a new equilibrium. The temperature response depends on both the magnitude of the initial forcing and the temperature-driven radiative feedbacks, which are represented by the climate feedback parameter (α) in equation (1).

The usefulness of radiative forcing as a predictor of GSAT change is dependent on the climate feedback parameter being the same across different forcing agents or equivalently that the "efficacy" of a forcing agent is close to unity. We define efficacy as the relative change in GSAT induced by a forcing agent compared to carbon dioxide, for an equivalent perturbation in global mean energy balance at the top of the atmosphere (TOA) or tropopause. Previous studies have indicated that the efficacies of different forcing agents may vary significantly from unity (Hansen et al., 2005; Manabe & Stouffer, 1980; Marvel et al., 2015; Modak et al., 2016; Modak et al., 2018; Modak & Bala, 2019; Shindell, 2014; Shindell et al., 2015; Stuber et al., 2005). Hansen et al. (2005) showed that in a single model, efficacies varied widely across many forcing agents, identifying an efficacy of ~1.5 for methane and ~0.6 for black carbon (BC). Several studies have also found that solar forcing has an efficacy of less than one (Forster et al., 2000; Hansen et al., 2005; Modak et al., 2016; Schmidt et al., 2012).

An important factor to consider is that efficacies are dependent on the method by which radiative forcing is defined (Hansen et al., 2005). Traditionally, radiative forcing was calculated at the tropopause as an instantaneous radiative forcing (IRF) or by accounting for stratospheric temperature adjustment (RF; Forster et al., 2007; Ramaswamy et al., 2001). However, these methods do not account for rapid adjustments which occur in the troposphere, such as changes to temperature, water vapor, and clouds (Andrews & Forster, 2008; Colman & McAvaney, 2011; Gregory & Webb, 2008; Kamae & Watanabe, 2013). The Intergovernmental Panel on Climate Change Fifth Assessment Report were able to estimate effective radiative forcing (ERF) for a few forcing agents, where ERF accounts for rapid adjustments of both stratospheric and tropospheric parameters (Boucher et al., 2013; Myhre et al., 2013). The ERF framework provides a more complete understanding of the energy budget response to different drivers than more traditional radiative forcing definitions (Hansen et al., 2005; Sherwood et al., 2015) and accounts for some of the differences between GSAT responses to different drivers that were previously accounted for by employing nonunity efficacies. However, some studies using ERF still find efficacies deviating from unity (Marvel et al., 2015; Modak et al., 2018, 2016; Modak & Bala, 2019; Shindell et al., 2015). This is further complicated by the fact that there are various methods for calculating ERF in models, which could affect efficacies (Forster et al., 2016; Sherwood et al., 2015).

Recent studies using the ERF framework to analyze transient historical simulations suggest that tropospheric aerosols drive a stronger transient climate response than carbon dioxide by around 50% (Shindell, 2014; Shindell et al., 2015). This has been mainly attributed to the inhomogeneous nature of aerosol forcing, which over the historical period has been more concentrated in the Northern Hemisphere extratropics, where there is evidence from models for a stronger sensitivity of temperature to localized forcing (Shindell, 2014; Shindell et al., 2015). As historical aerosol forcing is negative, this suggests that the higher efficacy could in part account for a lower than expected observed GSAT change and climate sensitivity derived from the Earth's energy budget (Forster, 2016; Knutti et al., 2017; Shindell et al., 2012). Accounting for different forcing efficacies could significantly increase climate sensitivity estimates from the historical period (Kummer & Dessler, 2014; Marvel et al., 2015). Therefore, quantifying forcing efficacies and understanding how robust they are across different models are of significant importance.

Only the Goddard Institute for Space Studies (GISS) modeling center has systematically diagnosed forcings and efficacies from their climate model (Hansen et al., 2005; Marvel et al., 2015). The lack of results from other models makes it difficult to test the robustness of the earlier findings and determine the implications for the real world. In this study we make use of the Precipitation Driver and Response Model Intercomparison Project (PDRMIP) data set (Myhre et al., 2017) to perform a systematic multimodel analysis of forcing efficacies. We compute efficacies from idealized single forcing simulations for a range of drivers across 11 PDRMIP models. In addition, we examine how the location of aerosol forcing affects efficacies. We compute radiative forcing using multiple methods to better understand the robustness of earlier findings and the dependence of efficacy calculations on radiative forcing definition as the method of radiative forcing calculation with the least efficacy variation could become the determining factor in defining forcing (Hansen et al., 2005).



2. Methods

2.1. Models and Simulations

We analyze output from 11 global climate models (see supporting information Table S1 for model details) participating in PDRMIP (Myhre et al., 2017). PDRMIP simulations are designed as idealized single forcing experiments to probe the multimodel robustness of responses to individual forcings and understand the role of rapid adjustments. Simulations were performed for five core global single forcing perturbations: doubling CO_2 concentration (2×CO2), tripling methane concentration (3×CH4), five times sulfate aerosol concentrations or emissions ($5 \times SO4$), ten times BC aerosol concentrations or emissions ($10 \times BC$), and a two-percent increase in insolation (2%SOL). An additional nine simulations were performed by a subset of models, to investigate the effect of regional forcings and further forcing agents, as detailed in Table S2. The aerosol and ozone forcing experiments have spatially varying concentrations. The perturbations are made relative to either present-day or pre-industrial baseline conditions (see Table S1). Simulations were performed with a coupled ocean for 100 years. Two models (ECHAM-HAM and CESM-CAM4) used a slab ocean, and these results are excluded from multimodel mean calculations of response as they do not allow for ocean circulation changes. An additional set of 15 years of simulations were performed with sea surface temperatures and sea-ice fixed (fSST) at the present-day or pre-industrial model climatology in atmospheric and land surface versions of the coupled and slab models to estimate forcings. All responses are calculated by subtracting the control run from the corresponding perturbed run.

Note that the experiments were performed by models that both prescribe concentrations of greenhouse gases and aerosols and models that prescribe greenhouse gases but use emissions derived aerosol concentrations (Table S1). In model configurations that perturbed emissions, the atmospheric distribution of aerosols is determined by the models own aerosol and cloud schemes, which introduces an additional element of uncertainty that is not present in the experiments in which atmospheric concentrations are prescribed. The models also adopt different treatments of aerosol microphysics and/or stratospheric water vapor oxidation (Table S1). None of the models considered include full interactive chemistry. One model (MPI-ESM) did not include aerosols and hence did not perform the aerosol perturbation experiments. CESM-CAM4 and GISS excluded aerosol microphysics changes to clouds (the Twomey effect). Overall, the differing model setups mean that aerosol concentrations, and hence forcings, varied across the suite of models and this should be kept in mind when evaluating the spread of model responses.

2.2. Forcing and Response

The energy balance of the climate system can be described using a simple linear equation relating radiative forcing and response (e.g., Boer & Yu, 2003; Gregory, 2004; Hansen et al., 1981):

$$N = F - \alpha \Delta T, \tag{1}$$

where the net radiative imbalance at the TOA (*N*) due to a driver is equal to the radiative forcing (*F*) minus the radiative response ($\alpha \Delta T$) which scales with GSAT change (ΔT) and " α " denotes the climate feedback parameter in W m⁻²/K.

2.3. Forcing Definitions

There are various commonly used radiative forcing definitions which depend on the method used for separating forcing and response. In this study we compute six different radiative forcing estimates, as detailed below, which each account for different degrees of atmospheric and land surface adjustment.

2.3.1. Instantaneous Radiative Forcing (IRF_{toa} and IRF_{trop})

IRF at the TOA (IRF_{toa}) is calculated for the five core PDRMIP experiments. IRF is also calculated at the tropopause (IRF_{trop}) for 2×CO2 and 3×CH4. IRF_{toa} is calculated using the difference of the perturbed and base runs using a "double-call" to the radiation code for 5×SO4 (available in three models) and 10×BC (six models). In each case an additional call to the radiation code is performed with the aerosol species of interest removed, and the IRF_{toa} is taken as the difference of the full-aerosol and no-aerosol calls in the perturbed run minus the difference of the full-aerosol and no-aerosol calls in the control run. For 2×CO2 and 3×CH4, IRF_{toa} are estimated by substituting each model's control climatology into



the SOCRATES radiation code (Edwards & Slingo, 1996) and taking the difference of radiative fluxes at the TOA or tropopause between the perturbed and control greenhouse gas concentrations in each model. For 2%SOL, IRF_{toa} is calculated as $2\% \times SC \times (1 - A)$ /4, where *SC* is each model's solar constant and *A* is the control run planetary albedo.

2.3.2. Stratospheric Temperature Adjusted Radiative Forcing (RF_{strat})

The stratospheric temperature adjusted radiative forcing is calculated by subtracting all rapid adjustments from ERF_{sst} (described below) except for stratospheric temperature change. The individual rapid adjustments are computed using the radiative kernel method outlined by Smith, Kramer, et al. (2018) and account for adjustments in tropospheric temperature, surface temperature, water vapor, surface albedo, and clouds.

2.3.3. Fixed SST Effective Radiative Forcing (ERF_{sst})

We compute ERF_{sst} as the change in TOA radiative flux between the perturbed and control fSST simulations (Hansen et al., 2005). In these simulations SST-driven radiative feedbacks are inhibited. ERF_{sst} therefore includes rapid adjustments of the troposphere, stratosphere, and land surface.

2.3.4. Adjusted Effective Radiative Forcing (ERF_{ssta})

 ERF_{ssta} is an "adjusted" version of ERF_{sst} which accounts for the land surface temperature changes that occur in the fSST simulations. The effect of land surface temperature change on the TOA energy balance is computed using radiative kernels and subtracted from ERF_{sst} . An alternative method to account for the land surface temperature change is to adjust ERF_{sst} based on the climate feedback parameter computed from the coupled runs (Sherwood et al., 2015). This method assumes that the climate feedback parameter is the same when only land temperatures are increasing. Here both methods produce similar results, so only the kernel results are shown for brevity.

2.3.5. Regression Effective Radiative Forcing (ERF_{reg})

Following equation (1), ERF_{reg} is calculated using linear regression of global mean TOA flux (*N*) change against GSAT change (ΔT) relative to the baseline simulation for the first 20 years of the coupled simulations. ERF_{reg} is given by the intercept of the regression line where $\Delta T = 0$ (Gregory, 2004).

2.4. Efficacy Computation

2.4.1. Efficacy Based on Forcing

We compute forcing efficacy by dividing the temperature response in the coupled PDRMIP simulations by the radiative forcing, as defined above, and normalizing with respect to the CO₂ response:

$$E_f = \frac{\Delta T/F}{\Delta T_{2\times CO2}/F_{2\times CO2}},\tag{2}$$

where E_f is the efficacy of a given forcing agent, ΔT is the GSAT response to the global mean forcing (*F*), and $\Delta T_{2\times CO2}$ is the GSAT response to CO₂ global mean forcing ($F_{2\times CO2}$). The GSAT responses are taken as the mean difference between the perturbed and baseline coupled PDRMIP simulations for years 81–100 of simulations. We calculate efficacies using the six different forcing definitions described above, denoting the efficacies E_{irf_toa} , E_{irf_trop} , E_{rf_strat} , E_{erf_ssta} , E_{erf_ssta} , and E_{erf_reg} (corresponding to the forcing definitions IRF_{toa} , IRF_{trop} , RF_{strat} , ERF_{ssta} , and ERF_{reg} , respectively).

This definition of efficacy shows the effectiveness of climate drivers at changing GSAT on a centennial timescale similar to Hansen et al. (2005). We adopt this approach as the derived efficacies are more suited to analysis of long-term historical temperature change, which is the focus of this work. However, it differs from the methods used in some studies where efficacies are computed from transient historical simulations (Marvel et al., 2015; Shindell, 2014). We also tested a calculation of efficacies more akin to equilibrium conditions by estimating the long-term equilibrium temperature response from the flux imbalance toward the end of the model integration. Such efficacies differed by less than 0.05 compared to those presented here (or less than 0.1 for $10 \times BC$ integrations), with no clear systematic difference.

For each model, 5–95% confidence intervals are calculated based on the control run interannual variability. Confidence intervals (CIs) for ERF and ΔT are computed using equation (3):



$$CI = \frac{s}{\sqrt{n}} \times t_{n-1} \times \sqrt{2},$$
(3)

where *s* is the standard deviation of the annual means in the control run for the given variable, *n* is the number of years, t_{n-1} is the corresponding *t* value from the *t* table with n-1 degrees of freedom. The ERF and ΔT confidence intervals are then propagated through equation (2).

2.4.2. Efficacy Based on Feedback Parameter

In addition to the calculation of efficacies based on GSAT and radiative forcing, we also compute forcing efficacies using the climate feedback parameter. The climate feedback parameter is calculated using linear regression of global mean TOA radiative flux change against GSAT change in the coupled PDRMIP simulations. A higher feedback parameter is associated with a smaller climate sensitivity, as less GSAT change is required to restore the climate system to equilibrium. Forcing efficacies based on the climate feedback parameter can then be computed using equation (4):

$$E_{\alpha} = \frac{\alpha_{\rm CO2}}{\alpha_f},\tag{4}$$

where E_{α} is the forcing efficacy, α_{CO2} is the climate feedback parameter in response to CO₂, and α_f is the climate feedback parameter in response to a given forcing. This efficacy definition implicitly assumes that the climate feedback parameter is constant through time.

2.5. Simple Model

We construct a simple impulse response model to estimate historical GSAT changes based on TOA forcing and PDRMIP GSAT response curves for the different drivers. The GSAT responses across the duration of the PDRMIP integrations were normalized by ERF_{sst} for each model and the multimodel mean computed. Following Larson and Portmann (2016), the normalized GSAT responses (*R*) are well fit using a two-term exponential decay function of the form:

$$R(t) = A \left(1 - e^{-t/\tau_1} \right) + B \left(1 - e^{-t/\tau_2} \right),$$
(5)

where *A* and *B* are amplitudes, *t* is time, and τ_1 and τ_2 are time constants. The GSAT change over the historical period due to a given forcing agent can then be calculated using

$$\Delta T_{i} = \sum_{j=0}^{l} R_{i-j} (F_{j} - F_{j-1}), \tag{6}$$

where ΔT_i is the change in GSAT in year *i*, F_j and F_{j-1} are the global mean TOA forcing in year *j* and *j* - 1, and R_{i-j} is the discretized response curve where t = i - j. The total historical GSAT change time series is then computed using equation (6) for each historical forcing and summing the results.

Equation (6) treats each annual forcing change as initiating a separate GSAT impulse response going forward in time and then sums the responses together with those from past years to give a GSAT change through time associated with a given forcing agent. Distinct GSAT response curves are used for historical forcing from CO_2 , CH_4 , insolation, and aerosols (attributed 5×SO4 response curve). Since PDRMIP did not perform a volcanic perturbation experiment, the effect of volcanic aerosol is attributed to the 2% SOL response curve. All remaining historical forcings are attributed the 2×CO2 response curve. The historical forcing time series is an updated version of data from Myhre et al. (2013), see Dessler and Forster (2018) for details (http://climexp.knmi.nl/selectindex.cgi#ERF). The inclusion of volcanic episodic forcing within an impulse response framework is known to be imperfect (Gregory et al. 2016). Volcanoes are included here only to more or less represent the historic temperature trend in the impulse response model; their efficacy impacts are not examined in this work. Nevertheless, our multiforcing impulse response approach represents a considerable advance over previous work that assumed that all radiative forcings generally have the same global temperature response as CO_2 (Good et al., 2013; Smith, Forster, et al., 2018).



3. Results and Discussion

3.1. Radiative Forcing of Core PDRMIP Experiments

We first compare the different radiative forcing definitions across the five core PDRMIP experiments (Figure 1 and Table 1). For the multimodel mean, the three ERF definitions are broadly consistent in magnitude. ERF_{ssta} is systematically slightly greater in magnitude than ERF_{sst} and ERF_{reg} . Adjusting ERF_{sst} to account for the radiative feedbacks driven by land surface temperature change does not bring the results into closer agreement with ERF_{reg} , as noted in previous studies (Hansen et al., 2005; Sherwood et al., 2015).

 IRF_{toa} is considerably smaller in magnitude than ERF for 2×CO2 and 5×SO4 and greater in magnitude for 2%SOL and 10×BC. These differences are a result of rapid adjustments of the stratosphere and troposphere (detailed in Smith, Kramer, et al., 2018), which are accounted for in ERF. For 2×CO2, stratospheric temperature adjustment is the main factor driving the larger ERF, and therefore, the stratospheric temperature adjusted RF_{strat} is in close agreement with ERF for 2×CO2. Tropospheric adjustments have a greater effect on ERF for 2%SOL and 10×BC. Large adjustments in tropospheric temperature and clouds result in a much smaller 10×BC ERF than RF_{strat} (Smith, Kramer, et al., 2018; Stjern et al., 2017).

The intermodel spread in radiative forcing is largest for the aerosol perturbations, which is partly due to the mix of concentration- and emission-based perturbations, with emissions-based scaling factors giving a range of concentrations, depending on the respective aerosol lifetimes in the models. HadGEM3 is a significant outlier for 5×SO4 with ERF ranging from -7.8 to -8.5 W m⁻² depending on the calculation method. The 2%SOL and 3×CH4 experiments exhibit the smallest intermodel spread in radiative forcing. It should be noted that 3×CH4 and 10×BC produce smaller radiative forcing values than the other experiments, which increases the uncertainty of the calculated efficacies.

The *ERF*_{sst} spatial pattern differs significantly between the different forcing agents (Figure 2), which could affect the efficacies (Forster et al., 2000; Hansen et al., 2005; Shindell, 2014). The forcing due to 2×CO2 and 3×CH4 is fairly homogeneous across the globe, with a slightly greater magnitude in the tropics. 2% SOL forcing exhibits a larger meridional gradient, with stronger forcing in the tropics. The aerosol perturbations produce more inhomogeneous forcing patterns. Due to the short atmospheric lifetime of aerosols, the forcing is mainly concentrated in the Northern Hemisphere close to the emission sources. For the aerosol emission-based models, $5\times$ SO4 *ERF*_{sst} is more hemispherically asymmetric, with the forcing more strongly concentrated in the Northern Hemisphere (Figure S1). The 10×BC forcing is mainly concentrated over Africa and southern Asia, and the $5\times$ SO4 forcing is strongest over southern Asia, Europe, the North Pacific, and North Atlantic. *ERF*_{sst} is spatially similar for the emission- and concentration-based models (Figure S1).

3.2. Efficacies Based on IRF and RF_{strat}

The global mean efficacies for $3\times$ CH4, $5\times$ SO4, $10\times$ BC, and 2%SOL are shown in Figure 3 and Table 1. There is considerable variation between the different efficacy definitions. Both E_{irf_toa} and E_{irf_trop} vary considerably between the different forcing agents. E_{irf_toa} is substantially less than 1 for $3\times$ CH4, 2%SOL, and $10\times$ BC, with multimodel mean values of 0.48, 0.44, and 0.27, respectively, and relatively higher for $5\times$ SO4 with a very large model spread, ranging from 0.51 to 2.5. E_{irf_trop} is close to one for $3\times$ CH4 and 2%SOL but has multimodel mean values of 0.55 for $10\times$ BC and 3.0 for $5\times$ SO4, with considerable model spread. Hansen et al. (2005) similarly found BC to have a low efficacy when based on IRF (at the tropopause) but found methane to have an efficacy greater than 1. Consistent with Hansen et al. (2005), we find that across the PDRMIP models IRF is a poor predictor of the GSAT response to different forcing agents whether computed at the TOA or tropopause.

3.3. Efficacies Within the ERF Framework

When computed using ERF, multimodel mean efficacies (E_{erf_sst} , E_{erf_ssta} , and E_{erf_reg}) are generally much closer to unity across the different perturbation experiments (Figure 3 and Table 1). This demonstrates that ERF is a better predictor of the climate response to different forcing agents. Under the IRF framework, short timescale tropospheric and stratospheric adjustment processes are considered part of the climate response and thus affect the efficacies. In contrast, efficacies based on ERF only describe differences in feedback strength to perturbations, as fast adjustments are accounted for in the forcing. The high multimodel mean



Figure 1. *IRF*_{trop}, *IRF*_{toa}, *RF*_{strat}, *ERF*_{reg}, *ERF*_{ssta}, and *ERF*_{ssta} for the five core PDRMIP forcings. Colored diamonds denote the multimodel mean, and error bars show the intermodel standard deviation. Individual model results are represented by gray diamonds. Hollow gray diamonds denote models which perturbed emissions rather than concentrations in the aerosol experiments. One model (HadGEM3) is off the bottom of the scale for 5×SO4.

Table 1

PDRMIP Multimodel Mean Radiative Forcing (See section 2.3 for Definitions), GSAT Response, Climate Feedback Parameter Calculated Using First 20 years (α_2 20), Final 80 years (α_2 80), and Full 100 years (α) of Simulations, and Efficacies (See section 2.4 for Definitions) for the Five Core Experiments

	2×CO2	3×CH4	2%SOL	5×SO4	10×BC
$IRF_{trop} (W m^{-2})$	4.45 ± 0.11	1.04 ± 0.09	_	_	_
IRF_{toa} (W m ⁻²)	2.20 ± 0.22	1.13 ± 0.13	4.81 ± 0.05	-1.73 ± 1.02	2.17 ± 1.08
RF_{strat} (W m ⁻²)	3.76 ± 0.27	1.21 ± 0.28	4.55 ± 0.06	-3.64 ± 1.20	2.29 ± 1.15
ERF_{sst} (W m ⁻²)	3.71 ± 0.30	1.15 ± 0.25	4.17 ± 0.13	-3.71 ± 1.94	1.18 ± 0.75
$ERF_{ssta} (W m^{-2})$	4.19 ± 0.35	1.25 ± 0.31	4.32 ± 0.12	-3.78 ± 1.99	1.25 ± 0.80
$ERF_{reg} (W m^{-2})$	3.81 ± 0.57	1.06 ± 0.25	3.89 ± 0.29	-3.70 ± 1.79	0.83 ± 0.53
$\Delta T(\mathbf{K})$	2.44 ± 0.75	0.67 ± 0.33	2.46 ± 0.97	-2.45 ± 1.85	0.74 ± 0.54
$\alpha (W m^{-2}/K)$	-1.19 ± 0.50	-1.06 ± 0.53	-1.24 ± 0.51	-1.11 ± 0.34	-0.89 ± 0.46
$\alpha_{20} (W m^{-2}/K)$	-1.27 ± 0.52	-1.28 ± 0.57	-1.36 ± 0.51	-1.51 ± 0.85	-0.97 ± 0.77
$\alpha_{80} (W m^{-2}/K)$	-1.03 ± 0.57	-0.93 ± 0.73	-1.04 ± 0.51	-0.80 ± 0.36	-0.78 ± 0.60
$\Delta T/IRF_{trop} (\text{K/W m}^{-2})$	0.55 ± 0.19	0.61 ± 0.33	—	—	—
$\Delta T/IRF_{toa}$ (K/W m ⁻²)	1.14 ± 0.44	0.57 ± 0.35	0.52 ± 0.21	1.25 ± 0.81	0.25 ± 0.21
$\Delta T/RF_{strat}$ (K/W m ⁻²)	0.66 ± 0.24	0.57 ± 0.29	0.54 ± 0.21	0.66 ± 0.25	0.40 ± 0.25
$\Delta T/ERF_{sst}$ (K/W m ⁻²)	0.67 ± 0.22	0.58 ± 0.23	0.59 ± 0.23	0.62 ± 0.19	0.63 ± 0.37
$\Delta T/ERF_{ssta}$ (K/W m ⁻²)	0.59 ± 0.19	0.54 ± 0.22	0.57 ± 0.22	0.61 ± 0.18	0.60 ± 0.34
$\Delta T/ERF_{reg}$ (K/W m ⁻²)	0.66 ± 0.25	0.66 ± 0.32	0.63 ± 0.25	0.63 ± 0.22	1.22 ± 1.45
E _{irf_trop}		1.09 ± 0.35	0.89 ± 0.09	2.97 ± 1.86	0.54 ± 0.29
E _{irf_toa}	—	0.48 ± 0.17	0.44 ± 0.04	1.50 ± 0.99	0.27 ± 0.13
E_{rf_strat}	—	0.84 ± 0.21	0.81 ± 0.06	0.99 ± 0.09	0.55 ± 0.16
E_{erf_sst}	—	0.87 ± 0.15	0.87 ± 0.07	0.94 ± 0.16	0.87 ± 0.31
E _{erf_ssta}	_	0.91 ± 0.18	0.95 ± 0.07	1.04 ± 0.16	0.93 ± 0.32
E _{erf_reg}	_	0.97 ± 0.23	0.96 ± 0.09	0.95 ± 0.25	1.48 ± 1.09
E_{α}	_	1.22 ± 0.47	0.97 ± 0.12	1.01 ± 0.29	1.36 ± 0.26

Note. Uncertainty bounds are the standard deviation of the intermodel spread.



Figure 2. Multimodel mean *ERF*_{sst} distributions of core PDRMIP experiments. Hatching shows where the multimodel mean is less than the intermodel standard deviation.

 E_{erf_reg} of 1.48 for 10×BC can be mainly attributed to one outlying model, HadGEM3, which has an E_{erf_reg} of 4.1, largely due to a small ERF_{reg} of 0.18 W m⁻².

 ERF_{ssta} and ERF_{sst} are generally the best predictors (i.e., efficacy closest to unity) of the GSAT response to forcing across the PDRMIP models and forcing agents. The multimodel mean E_{erf_ssta} is systematically marginally closer to unity than E_{erf_sst} . This is due to the proportionately greater difference between ERF_{sst} and ERF_{ssta} for 2×CO2 compared to the other perturbations. The greater difference may be due to the physiological effect of CO₂ on vegetation. Increasing atmospheric CO₂ concentration reduces stomatal opening, which can reduce evapotranspiration and enhance surface warming (Andrews et al., 2010; Boucher et al., 2009; Cao et al., 2010; Field et al., 1995; Pu & Dickinson, 2014). As a result, CO₂ drives greater land surface warming than any other driver in the fSST simulations used to compute ERF_{sst} . Therefore, accounting for

100



Efficacy of Climate Forcings

land surface temperature-driven feedbacks in the fSST simulations using radiative kernels enables a slightly better prediction of the climate response to forcing.

Multimodel mean E_{erf_ssta} and E_{erf_sst} are both close to one for 5×SO4 (0.98 and 0.94, respectively) and slightly less than 1 for 10×BC (both 0.87). Marvel et al. (2015) similarly found that the efficacy of tropospheric aerosols is closer to one when using ERF, rather than IRF. However, the PDRMIP results are in contrast to previous work which suggests that tropospheric aerosols have an efficacy around 40–50% higher than CO_2 over the historical period (Rotstayn et al., 2015; Shindell, 2014; Shindell et al., 2015). The high efficacy of aerosols has mainly been attributed to the inhomogeneous pattern of the forcing (Shindell, 2014). Over the historical period tropospheric aerosol forcing has been concentrated in the Northern Hemisphere where the temperature response may be enhanced by strong positive feedbacks at high latitudes (Pithan & Mauritsen, 2014), less moisture availability over land (Joshi et al., 2008), and a faster transient response. However, despite the fact that the forcing is predominantly concentrated in the Northern Hemisphere in the 5×SO4 and 10×BC simulations (Figure 2), an enhanced GSAT response is not evident in most models.

 E_{erf_sst} for each model and experiment is shown in Figure 4, along with 5–95% confidence intervals. There is considerable intermodel spread in the 5×SO4 and 10×BC efficacies. For all concentration-based models, the efficacy of 5×SO4 is less than one (0.73–0.95), whereas the emission-based models exhibit efficacies both higher and lower than one (0.89–1.16). How aerosol emissions translate into atmospheric burden, and hence forcing, may introduce a larger degree of variability in efficacy results than indicated from the concentration-based experiments. Both concentration- and emission-based models produce a large intermodel spread in efficacies for 10×BC, ranging from 0.36 to 1.33 (Figure 4). The large uncertainties can mainly be attributed to a small TOA forcing and weak GSAT response (Stjern et al., 2017). In addition, both the forcing and climate response to BC are dependent on the vertical profile of the BC aerosols (Ban-Weiss et al., 2012); therefore, TOA ERF may not be a useful predictor of the GSAT response. Looking across the model ensemble as a whole there is the efficacy of sulfate or BC aerosols that varies substantially leaving it unclear whether it significantly differs from unity.

Figure 3. Global mean E_{irf_trop} , E_{irf_toa} , E_{rf_strat} , E_{erf_sst} , E_{erf_ssta} , and E_{α} calculated for PDRMIP forcing experiments. Colored diamonds denote the multimodel mean, and error bars show the intermodel standard deviation. Individual model results are represented by gray diamonds. Hollow gray diamonds denote models which perturbed emissions rather than concentrations in the aerosol experiments. One model result lies off the top of the scale for 5×SO4 E_{irf_trop} (MIROC-SPRINTARS) and 10×BC E_{erf_r} (HadGEM3).



Efficacy of Climate Forcings (E_{eff sst})

Figure 4. *E_{erf_sst}* estimates for each PDRMIP model. Error bars denote the 5–95% confidence interval based on the interannual variability of the control run. Models which perturbed emissions rather than concentrations in the aerosol experiments are shown in red. CESM1-CAM4 and ECHAM-HAM are run with a slab ocean.

Methane has an E_{erf_sst} of less than one in all models except one, ranging from 0.56 to 1.15. However, the uncertainty is generally large, and for seven of the 10 models, an efficacy of one lies within the confidence intervals. Hansen et al. (2005) found methane to have an efficacy close to unity (0.87–1.08) when based on ERF, but more recently a lower estimate of 0.81 was obtained (Modak et al., 2018). Modak et al. (2018) attribute the lower efficacy to differing rapid adjustments of upper tropospheric temperature, lower stratospheric temperature, and clouds, which then affect the slow temperature response and lapse rate feedback.

 E_{erf_sst} is slightly less than one across models for 2%SOL, ranging from 0.77 to 0.95. Previous studies generally find efficacies of less than one for solar forcing (Forster et al., 2000; Hansen et al., 2005; Marvel et al., 2015; Modak et al., 2016). The weaker efficacy may be due to the meridional structure of solar forcing (Figure 2), which leads to less warming at high latitudes (Figures 5 and 6). Warming in the tropics is more efficiently damped by radiative feedbacks (Kang & Xie, 2014). Modak et al. (2016) also suggest that tropospheric cloud adjustments and stratospheric warming lead to the weaker efficacy for solar forcing.

3.4. Spatial Pattern and Vertical Structure of Temperature Response

We next look at how the surface temperature changes are manifested spatially. Figure 5 shows the spatial pattern of surface temperature change normalized by GSAT change for each perturbation. Despite the large spatial differences in forcing patterns for the different drivers, the temperature responses project onto broadly similar spatial patterns, with greater changes occurring in the Northern Hemisphere at higher latitudes and over land regions. The hemispheric asymmetry is strongest for the aerosol perturbations, which is consistent with the forcing being concentrated in the Northern Hemisphere (Figure 2).

Figure 6 displays the surface temperature response normalized by ERF_{sst} for each perturbation relative to the 2×CO2 surface air temperature response. The larger hemispheric asymmetry in surface temperature change for the aerosol perturbations is evident. The temperature response is robustly weaker over most of the Southern Hemisphere for 5×SO4 and 10×BC, where the local forcing is comparatively small. In the Northern Hemisphere there are various regions where surface temperature is more sensitive to the aerosol perturbations. In response to 5×SO4, surface temperature is notably more sensitive over the Norwegian and Barents Seas than for 2×CO2. The higher temperature sensitivity in the northern midlatitudes for 5×SO4.



Figure 5. Multimodel mean surface temperature response for core PDRMIP forcing experiments (years 81–100 of coupled runs), normalized by the GSAT response. Hatching shows where the multimodel mean is less than the intermodel standard deviation.

appears to be more prominent in the emission-based models (Figure S2). However, given the different formulations and different aerosol concentrations within the individual emission-based models, it is difficult to determine what factors would lead to this effect.

In response to $10\times$ BC, surface temperature is more sensitive than for $2\times$ CO2 in western Asia, the northern Pacific, North America, and the Mediterranean. Interestingly, the regions of higher temperature sensitivity to the aerosol forcing do not necessarily occur in regions where the emissions and forcing are greatest, nor where they are comparatively larger than for $2\times$ CO2.

The $3\times$ CH4 perturbation leads to a very similar spatial pattern of warming to $2\times$ CO2. There are some locations over land where the $3\times$ CH4 temperature response is robustly weaker than for $2\times$ CO2, which coincide with the Amazon, central African, and boreal forests. In these densely vegetated regions, the physiological

100





Figure 6. Multimodel mean difference in normalized surface temperature responses for each of the core PDRMIP perturbation experiments relative to temperature response to $2 \times CO2$. Temperature responses are normalized by ERF_{sst} . Hatching shows where the multimodel mean is less than the intermodel standard deviation.

effect of CO_2 on plants can strongly reduce evapotranspiration and therefore enhance warming (Andrews et al., 2010; Cao et al., 2010). Modeling studies suggest that the physiological effect enhances land mean surface warming in response to CO_2 by around 15% (Boucher et al., 2009; Cao et al., 2010; Pu & Dickinson, 2014). This may also contribute to the robustly weaker temperature response to 2%SOL over many land regions relative to 2×CO2. The temperature response to 2%SOL is robustly weaker than for 2×CO2 at high latitudes in the Northern Hemisphere which is likely due to the weaker forcing there relative to the tropics for 2%SOL.

We now consider the vertical structure of the temperature responses in the PDRMIP experiments, as shown in Figure 7. Except for the 10×BC experiment, each of the forcings exhibits broadly similar large-scale patterns of tropospheric temperature change, with amplification in the tropical upper-troposphere and near-surface amplification at the poles, particularly in the Arctic. The amplified tropical upper tropospheric temperature change is a robust feature of the response to greenhouse gases in global climate models (Ceppi & Shepherd, 2017; Gillett et al., 2000; Rotstayn et al., 2013; Santer et al., 2013) with a similar trend also evident in observations since the late 1950s (Santer et al., 2017; Vinnikov et al., 1996). Radiative forcing in the tropics produces less surface heating than the poles as more energy is used to increase evaporation. This leads to increased latent heat release in the upper troposphere, and the vertical temperature profile adjusts toward the moist adiabatic lapse rate (Mitchell et al., 1990; Wilson & Mitchell, 1987).

The tropospheric temperature response to aerosols is less hemispherically symmetric, consistent with the asymmetric nature of the aerosol forcing (Figure 2) and surface temperature response (Figure 5). 10×BC in particular drives a noticeably different pattern of atmospheric temperature change to the other drivers, with strong heating near the tropopause. Unlike most other drivers of climate change, BC affects atmospheric temperature predominantly through rapid adjustments rather than through surface temperature mediated processes (Stjern et al., 2017). BC strongly absorbs shortwave radiation, directly heating the





Figure 7. PDRMIP multimodel zonal mean temperature response to forcing normalized by GSAT change. The gray contours show the multimodel mean zonal temperature climatology. The stippling shows where the MMM difference is less than ± 1 standard deviation of the intermodel spread.

atmosphere without strongly warming the surface, which can account for the qualitatively different vertical profile of warming.

In the stratosphere, $2\times$ CO2 is the only driver that induces radiative cooling across the globe. 2%SOL induces stratospheric heating at most latitudes owing to the increased absorption of ultraviolet radiation by ozone. The one exception is HadGEM3 which simulates a relatively large (~75%) increase in stratospheric water vapor in the 2%SOL experiment, which offsets the shortwave-driven stratospheric heating (not shown). In 3×CH4 and 5×SO4 the stratospheric temperature changes show regions of heating and cooling, which are indicative of changes in the large-scale stratospheric overturning circulation. However, most of the PDRMIP models do not fully resolve the stratosphere, which may limit their ability to capture changes in stratospheric circulation (Charlton-perez et al., 2013).

3.5. Regional Aerosol Perturbations

To investigate whether the location of aerosol forcing impacts on its efficacy, we analyze the three regional PDRMIP experiments: 10×BCasia, 10×SO4asia, and 10×SO4eur (see Table S2). Global mean ERF_{sst} results for the regional simulations are shown in Figure 8a and Table S3. As expected,10×BCasia produces strong forcing over Asia but globally averaged the forcing is small, meaning a robust efficacy calculation cannot be performed. 10×SO4asia produces a fairly large negative global mean ERF_{sst} ranging from -0.46 to -1.15 W m⁻², and 10×SO4eur produces a smaller negative ERF_{sst} ranging from -0.27 to -0.41 W m⁻².

Comparing the two regional sulfate perturbation experiments, there is no clear difference in efficacies across the models due to the location of the forcing (Figure 8b). E_{erf_sst} ranges from 0.66 to 1.33 for 10×SO4eur, and 0.75 to 1.13 for 10×SO4asia, showing a similarly large intermodel spread to the global 5×SO4 experiment. Liu et al. (2018) also noted that regional and global sulfate aerosol forcing drives a similar GSAT response to CO₂ forcing. Previous studies, mainly based on the GISS model, suggest that the concentration of

100



(a) Effective Radiative Forcing (ERF_{sst})

(b) Efficacy of Climate Forcing (E_{eff sst})



Figure 8. ERF_{sst} (top) and E_{erf_sst} (bottom) of each model for global and regional forcing experiments performed by a subset of PDRMIP models. The error bars denote the 5–95% confidence interval based on the interannual variability of the control run. Models which perturbed emissions rather than concentrations in the aerosol experiments are shown in red. CESM1-CAM4 and ECHAM-HAM are run with a slab ocean.

aerosol forcing at high latitudes may enhance its efficacy (Shindell et al., 2015; Shindell & Faluvegi, 2009). Kasoar et al. (2018) also found that the GSAT response to sulfate aerosol forcing was sensitive to the location of the source region. Similarly, in our analysis both GISS and HadGEM3 have higher efficacies for 10×SO4eur (1.32 and 1.33, respectively) than for 10×SO4asia (1.04 and 0.78, respectively). However, this result is not robust across the PDRMIP models, with no other model exhibiting a higher sensitivity to aerosol forcing over Europe than over Asia. In addition, the 10×SO4asia and 10×SO4eur efficacy results are not significantly different from unity for any of the models except NorESM. In five out of the seven models, concentrations were perturbed rather than emissions. Incorporating the models' own aerosol schemes to determine the atmospheric burden would likely introduce further uncertainty and model spread not seen in this analysis.

The two regional sulfate experiments drive a similar hemispheric asymmetry in surface temperature change (Figure 9) as seen for the global perturbation (Figure 5). Surface temperature responds more strongly across the Northern Hemisphere than the Southern Hemisphere in both 10×SO4asia and 10×SO4eur, even in





Figure 9. Multimodel mean ERF_{sst} for (a) 10×SO4asia and (b) 10×SO4eur. Multimodel mean surface temperature response normalized by the GSAT response for (c) 10×SO4asia and (d) 10×SO4eur. Multimodel mean difference in normalized surface temperature response relative to 2×CO2 for (e) 10×SO₄asia and (f) 10×SO₄eur. The temperature response is normalized by ERF_{sst} .

locations far from the forcing region. The strongest cooling occurs locally to the region of forcing. Similarly, Kasoar et al. (2018) found that when SO_2 emissions are removed from specific regions, the temperature response does tend to be strongest local to the forcing, but there is also a common broad pattern response regardless of forcing location. The particularly strong cooling over Europe for 10×SO4eur is consistent with high-latitude forcing being less effectively damped by radiation (Kang & Xie, 2014).

3.6. Efficacy of Other Forcing Agents

Using a subset of the PDRMIP models, a number of further forcing scenarios were analyzed to determine the efficacies of halocarbons, tropospheric ozone, land use change, and BC with a shorter lifetime (Figure 8 and Table S3). The spatial pattern of ERF_{sst} is shown in Figure S3. E_{erf_sst} for CFCs (CFC-12 and CFC-11) is generally close to or slightly less than one. For CFC-11 the models have an E_{erf_sst} slightly smaller than unity, with values ranging from 0.87 to 0.95. However, this smaller efficacy is not robust as the 5–95% uncertainty range for each model still includes unity. CFC-12 exhibits more model spread with E_{erf_sst} ranging from 0.69 to 1.09. E_{erf_sst} for N₂O ranges from 0.88 to 1.02, with only one model (HadGEM2) having an efficacy significantly different from one. This is consistent with Hansen et al. (2005) who also found that CFCs and N₂O have an efficacy close to one when computed using ERF.



Ozone forcing in NorESM and MIROC-SPRINTARS produces a global mean efficacy of 0.95 and 0.92, respectively, which is larger than previous results (Hansen et al., 2005; Marvel et al., 2015; Yoshimori & Broccoli, 2008). Spatially, the ozone forcing produces a more hemispherically asymmetric warming pattern than 2×CO2 with a higher sensitivity in the Northern Hemisphere (Figure S4). The land use change experiment produces a very small global mean ERF_{sst} ranging from -0.14 to 0.15 W m⁻² across models, resulting in highly uncertain efficacies. Shortening the lifetime of BC in GISS appears to make little difference to the efficacy. For ECHAM-HAM shortening BC lifetime produces a very small ERF_{sst} resulting in a highly uncertain efficacy.

3.7. Climate Feedback Parameter

It is possible to estimate forcing efficacies based on the climate feedback parameter (E_{α} in Figure 3 and Table 1), under the assumption that the climate feedback parameter remains constant in time. However, various studies have shown that the climate feedback parameter in response to an abrupt increase in CO₂ is not constant and tends to change such that the climate sensitivity increases as equilibrium is approached (Andrews et al., 2015; Armour et al., 2013; Gregory, 2004; Haugstad et al., 2017; Rose & Rayborn, 2016; Rugenstein et al., 2016; Winton et al., 2010). This is generally attributed to evolving patterns of surface temperature change, which actuate different local feedbacks on different timescales (Andrews et al., 2015, 2018; Rugenstein et al., 2016). It is therefore important to understand the time evolution of the climate feedback parameter and whether this is affected by the changing mix of time-evolving forcings (Marvel et al., 2015).

The climate feedback parameters computed using the first 20 years (α_20) and final 80 years (α_80) of the five core PDRMIP simulations are shown in Table 1. Across forcing experiments the feedback parameter is higher in the first 20 years and has a tendency to decrease through the runs in the majority of models. The time evolution of the feedback parameter for the 5×SO4 simulation is notably different to 2×CO2. The feedback parameter in response to sulfate forcing tends to exhibit a larger reduction in magnitude through the simulations, with a multimodel mean of -1.51 ± 0.85 W m⁻²/K for the first 20 years decreasing to -0.80 ± 0.36 W m⁻²/K in the final 80 years. This is in comparison to -1.27 ± 0.52 W m⁻²/K and -1.03 ± 0.57 W m⁻²/K for 2×CO2. Therefore, the effective climate sensitivity in terms of temperature change per unit forcing computed from the first 20 years of simulations is lower for sulfate than for CO₂. This is in contrast to recent studies which suggest that the temperature change per unit forcing of tropospheric aerosols could be higher than for CO₂ within transient simulations (Shindell, 2014; Shindell et al., 2015).

3.8. Impact of Efficacies on Historical Temperature Change

Figure 10a shows the multimodel mean GSAT response curves normalized by ERF_{sst} for 2×CO2, 5×SO4, and 2%SOL. It can be seen that in the early stages of the integrations the multimodel mean GSAT response to sulfate lags behind that of CO₂, before converging again later in the simulations. The lag in GSAT change is due to subdued warming in the Southern Hemisphere (Figure S5). The response curve for 10×BC (Figure S6) is similar to the 5×SO4 response, but with proportionately larger uncertainties owing to the overall smaller GSAT response to BC in the experiments. 2%SOL drives a smaller GSAT response than 2×CO2 throughout the simulations. This is due to relatively weaker warming in the Northern Hemisphere (Figure S5), which is a manifestation of the weaker polar amplification seen in Figure 6. The 3×CH4 response curve (Figure S6) is similar to the 2%SOL. The differences in the multimodel mean response curves for each forcing type, however, are substantially smaller than the intermodel spread (gray shading in Figure 10a shows the standard deviation of intermodel spread).

We construct a simple model to estimate GSAT change in response to historical TOA forcing based on the PDRMIP response curves (see section 2.5). To investigate the importance of efficacies for interpreting historical trends, we use the simple model to compute GSAT change using the same response curve (2×CO2) for all forcing agents (Figures 10b and 10c, black lines) as well as using distinct response curves for CO₂, CH₄, insolation, total aerosol (represented by the 5×SO4 response curve), and volcanic forcing (represented by the 2%SOL response curve). All other remaining forcings are attributed the 2×CO2 response curve. This is done using both the multimodel mean response curves and for each individual model (see Figure S7). Based on the multimodel mean response curves, accounting for the different multimodel mean efficacies has very little impact on the historical GSAT trend with the largest differences found following volcanic eruptions (Figures 10b and 10c, difference shown in blue).





(a) PDRMIP multi-model mean response curves

Figure 10. Panel (a) shows the multimodel mean GSAT response curves as a function of time for 2×CO₂, 5×SO₄, and 2% SOL normalized by *ERF_{sst}*. The intermodel standard deviation is shown in gray. Panel (b) shows the multimodel mean historical GSAT change computed using the impulse response model described in section 2.5 not accounting for efficacies (red) and accounting for efficacies (black dotted). The difference is shown in blue. Individual model results not accounting for efficacies are shown by dark gray lines GSAT and accounting for efficacies by light gray lines. Panel (c) shows the same as panel (b) for the period 1915–2014 and therefore includes no extrapolation of the response curves beyond the 100 years of PDRMIP data.

Accounting for efficacies has a larger effect for some individual models, in particular CanESM2. However, for many of the models, accounting for efficacies has a very small effect on the historical GSAT trend. Monte Carlo simulations (10,000 realizations) were performed to understand how uncertainty in the efficacy affects simulation of historic temperatures in the simple model. Each efficacy was randomly sampled across the PDRMIP model range for each forcing agent separately, and these sampled efficacies were used to scale the forcing-response curves in the simple model to construct an overall GSAT trend (see Figures 10b and 10c). The simple model gives a best estimate of GSAT change over 1915–2015 of 1.2 \pm 0.17 K, with the 5–95% range quoted due to the PDRMIP model efficacy variations. This GSAT trend was used to derive historic climate sensitivity using the approach taken in many papers (e.g., Forster, 2016; Marvel et al., 2015) of estimating α from equation (1) and equilibrium climate sensitivity (ECS) from $F_{\times CO2}/\alpha$. The 1.2 \pm 0.17 K GSAT trend estimate over 1915–2015 was combined with estimates of *F*, *N*, and $F_{\times CO2}$ of 2.2, 0.6, and 3.7 W m⁻²,



respectively, from Forster et al. (2016). This gives an ECS from the historical of 2.91 ± 0.4 K, with the 5% to 95% uncertainties due to possible efficacy variation of roughly $\pm 14\%$; this can be compared to a nonefficacy best estimate ECS of 2.88 K. The difference is considerably smaller than the 1.0 K uplift from 2.0 to 3.0 K suggested by Marvel et al. (2015), when ERF efficacy was accounted for their model.

Studies which have found significant differences in efficacies (Marvel et al., 2015; Shindell, 2014) which could affect climate sensitivity have tended to be based on historical transient simulations, rather than step-change experiments as used here. The differing methodologies of computing forcing and efficacies could contribute to differing results (Lewis and Curry, 2018). In particular, the version of the GISS model used here had no aerosol indirect effects or interactive chemistry so is not directly comparable to some earlier analyses. It should be noted that the aerosol perturbations used in this study were very large, and it is assumed that the temperature response scales linearly with forcing strength, so that the efficacies computed are applicable for the historical period. Further studies would be needed to explore this, and long model integrations would likely be needed to improve the signal to noise compared to that found here and in earlier studies.

4. Conclusions

Quantifying the effectiveness of different forcing agents in driving surface temperature change is important for understanding the real-world climate sensitivity and for projecting future levels of surface warming for different emission pathways. In this study we have presented a systematic multimodel analysis of the efficacies of a wide range of different climatic drivers, in particular methane, insolation, sulfate, and BC aerosols. Consistent with previous studies, we find that efficacies calculated using IRF vary significantly across drivers. This is because rapid atmospheric adjustments to both CO_2 (e.g., stratospheric temperatures) and other drivers (e.g., cloud adjustments) are not captured by the IRF concept and are instead attributed as part of the long-term climate response. When computed based on ERF, which does account for rapid adjustments, the multimodel mean efficacies are much closer to unity across drivers. However, there remains a large intermodel spread, in particular in response to sulfate and BC forcing.

We find that ERF definitions based on fixed SST experiments (ERF_{sst} and ERF_{ssta}) are the best predictors of the climate response (efficacies closest to one) across different forcing agents. Efficacies are marginally closer to one when ERF_{sst} is adjusted to take into account land surface temperature change using radiative kernels. E_{erf_sst} is less than one across all models for solar forcing (0.77–0.95). Multimodel mean E_{erf_sst} is slightly less than one for methane, sulfate, and BC forcing, but there is a greater intermodel spread (3×CH4: 0.56–1.15, 5×SO4: 0.73–1.16, and 10×BC: 0.36–1.33). Spatially, there are some differences in the surface temperature response to different forcings. The aerosol experiments exhibit weaker temperature changes in the Southern Hemisphere and stronger responses in some regions of the Northern Hemisphere relative to CO₂. However, the regions where surface temperature is more sensitive to aerosol forcing are *not* generally where the forcing is strongest. Sulfate forcing in Europe has a slightly higher efficacy than forcing in Asia in the multimodel mean. However, the model spread is large, and there is no robust indication across models that the efficacies differ from unity. The considerable model spread in aerosol efficacies found in this study indicates that previous findings on aerosol efficacies relying on single model studies may not be robust.

The radiative feedbacks in response to GSAT change are generally not constant and evolve through time. Across the five core experiments the climate feedback parameter tends to decrease through the simulations (i.e., sensitivity increases). This agrees with previous work on the evolution of model response to CO_2 (e.g., Armour, 2017). The time evolution of the climate feedback parameter differs slightly between forcing experiments. In response to sulfate forcing the change in the climate feedback parameter is particularly large. The GSAT response to 5×SO4 initially lags behind the response to 2×CO2 due to a subdued southern hemispheric response. However, as the climate feedback parameter reduces later in the simulation, GSAT begins to change faster in response to sulfate forcing than for CO_2 .

Despite evidence of a forcing-response relationship that depends on forcing type, we find that overall there is not a robust indication that accounting for the different GSAT response curves gives an appreciably different GSAT evolution over the historical period than assuming that the response curves of all forcing agents follow that of CO₂. This result is valid for forcing-response curves normalized by ERF. Forcing definitions that do



Acknowledgments

The PDRMIP model output is publicly available (for data access, visit http:// www.cicero.uio.no/en/PDRMIP/ PDRMIP-data-access), T. B. R. and P. M. F. were supported by Natural Environment Research Council Grant NE/N006038/1. P. M. F., A. C. M., G. M., B. H. S., O. B., C. J. S., and T. A. acknowledge support from the European Union's Horizon 2020 Research and Innovation Programme under Grant Agreement 820829 (CONSTRAIN). A. C. M. was supported by a NERC Independent Research Fellowship (NE/M018199/1). T. A. was supported by the Joint UK BEIS/Defra Met Office Hadley Centre Climate Programme (GA01101). B. H. S., G. M., and \emptyset . H. were funded by the Research Council of Norway, through the grant NAPEX (229778). D. S. and G. F. thank NASA GISS for funding and acknowledge the NASA High-End Computing Program through the NASA Center for Climate Simulation at Goddard Space Flight Center for computational resources. O. B. acknowledges HPC resources from TGCC under the gencmip6 allocation provided by GENCI (Grand Equipement National de Calcul Intensif). T. T. is supported by the NEC SX-ACE supercomputer system of the National Institute for Environmental Studies, Japan, the Environmental Research and Technology Development Fund (S-12-3) of the Ministry of Environment, Japan, and JSPS KAKENHI Grants JP15H01728 and JP15K12190. M. K., D. S., and A. V. were supported by the Natural Environment Research Council under Grant NE/K500872/1 and from the Grantham Institute at Imperial College. Simulations with HadGEM2 and HadGEM3-GA4 were performed using the MONSooN system, a collaborative facility supplied under the Joint Weather and Climate Research Programme, which is a strategic partnership between the Met Office and the Natural Environment Research Council. D. O. and A. K. were supported by the Norwegian Research Council through the projects EVA (229771), EarthClim (207711/E10), NOTUR (nn2345k), and NorStore (ns2345k). D. W. P. and P. S. acknowledge funding from Natural Environment Research Council Projects NE/J022624/1 (GASSP), NE/L01355X/1 (CLARIFY), and NE/P013406/1 (A-CURE). P. S. also acknowledges funding from the European Research Council project RECAP under the European Union's Horizon 2020 Research and Innovation Programme with Grant Agreement 724602. The ECHAM6-HAM2 simulations were performed using the ARCHER UK National

not account for rapid adjustments do exhibit considerable efficacy variation. For example, using RF_{strat} to construct historic GSAT would overestimate the role of BC if efficacies were not accounted for.

Our results indicate that ERF efficacies could be close enough to one that they will not significantly affect climate sensitivity estimates derived from the historical period temperature trend. However, considerable uncertainty in our computed efficacies remains. A role for efficacy variation in historical temperature evolution (of $\pm 14\%$) cannot be ruled out, with a similar relative effect on derived estimates of ECS (± 0.4 K), indicating that it could still be important to consider. Future work should focus on understanding the physical differences for the model spread to determine where variations are robust or more an issue of signal to noise.

References

- Andrews, T., Doutriaux-Boucher, M., Boucher, O., & Forster, P. M. (2010). A regional and global analysis of carbon dioxide physiological forcing and its impact on climate. *Climate Dynamics*, 36(3–4), 783–792. https://doi.org/10.1007/s00382-010-0742-1
- Andrews, T., & Forster, P. M. (2008). CO₂ forcing induces semi-direct effects with consequences for climate feedback interpretations. *Geophysical Research Letters*, 35, L04802. https://doi.org/10.1029/2007GL032273
- Andrews, T., Gregory, J. M., Paynter, D., Silvers, L. G., Zhou, C., Mauritsen, T., et al. (2018). Accounting for changing temperature patterns increases historical estimates of climate sensitivity. *Geophysical Research Letters*, 45, 8490–8499. https://doi.org/10.1029/2018GL078887 Andrews, T., Gregory, J. M., & Webb, M. J. (2015). The dependence of radiative forcing and feedback on evolving patterns of surface
- temperature change in climate models. Journal of Climate, 28(4), 1630–1648. https://doi.org/10.1175/JCLI-D-14-00545.1
- Armour, K. C. (2017). Energy budget constraints on climate sensitivity in light of inconstant climate feedbacks. *Nature Climate Change*, 7(5), 331–335. https://doi.org/10.1038/nclimate3278
- Armour, K. C., Bitz, C. M., & Roe, G. H. (2013). Time-varying climate sensitivity from regional feedbacks. Journal of Climate, 26(13), 4518–4534. https://doi.org/10.1175/JCLI-D-12-00544.1
- Ban-Weiss, G. a., Cao, L., Bala, G., & Caldeira, K. (2012). Dependence of climate forcing and response on the altitude of black carbon aerosols. *Climate Dynamics*, *38*(5–6), 897–911. https://doi.org/10.1007/s00382-011-1052-y
- Boer, G., & Yu, B. (2003). Climate sensitivity and response. *Climate Dynamics*, 20(4), 415–429. https://doi.org/10.1007/s00382-002-0283-3
 Boucher, O., Jones, A., & Betts, R. A. (2009). Climate response to the physiological impact of carbon dioxide on plants in the Met Office Unified Model HadCM3. *Climate Dynamics*, 32(2–3), 237–249. https://doi.org/10.1007/s00382-008-0459-6
- Boucher, O., Randall, D., Artaxo, P., Bretherton, C., Feingold, G., Forster, P., et al. (2013). Clouds and aerosols. In Climate change 2013: The physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 571–657). https://doi.org/10.1017/CBO9781107415324.016
- Cao, L., Bala, G., Caldeira, K., Nemani, R., & Ban-Weiss, G. (2010). Importance of carbon dioxide physiological forcing to future climate change. Proceedings of the National Academy of Sciences, 107(21), 9513–9518. https://doi.org/10.1073/pnas.0913000107
- Ceppi, P., & Shepherd, T. G. (2017). Contributions of climate feedbacks to changes in atmospheric circulation. *Journal of Climate*, 30(22), 9097–9118. https://doi.org/10.1175/JCLI-D-17-0189.1
- Charlton-perez, A. J., Baldwin, M. P., Birner, T., Black, R. X., Butler, A. H., Calvo, N., et al. (2013). On the lack of stratospheric dynamical variability in low-top versions of the CMIP5 models. *Journal of Geophysical Research: Atmospheres*, 118, 2494–2505. https://doi.org/ 10.1002/jgrd.50125
- Colman, R. a., & McAvaney, B. J. (2011). On tropospheric adjustment to forcing and climate feedbacks. *Climate Dynamics*, 36(9–10), 1649–1658. https://doi.org/10.1007/s00382-011-1067-4
- Dessler, A. E., & Forster, P. M. (2018). An estimate of equilibrium climate sensitivity from interannual variability. Journal of Geophysical Research: Atmospheres, 123, 8634–8645. https://doi.org/10.1029/2018JD028481
- Edwards, J. M., & Slingo, A. (1996). Studies with a flexible new radiation code. I: Choosing a configuration for a large-scale model. Quarterly Journal of the Royal Meteorological Society, 122(531), 689–719. https://doi.org/10.1256/smsqj.53106
- Field, C. B., Jackson, R. B., & Mooney, H. A. (1995). Stomatal responses to increased CO₂: Implications from the plant to the global scale. *Plant, Cell and Environment, 18*(10), 1214–1225. https://doi.org/10.1111/j.1365-3040.1995.tb00630.x
- Forster, P., Ramaswamy, V., Artaxo, P., Berntsen, T., Betts, R., Fahey, D.W., et al. (2007). Changes in atmospheric constituents and in radiative forcing. *Climate change 2007*: The physical science basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. https://doi.org/10.1103/PhysRevB.77.220407
- Forster, P. M. (2016). Inference of climate sensitivity from analysis of Earth's energy budget. Annual Review of Earth and Planetary Sciences, 44(1), 85–106. https://doi.org/10.1146/annurev-earth-060614-105156
- Forster, P. M., Richardson, T., Maycock, A. C., Smith, C. J., Samset, B. H., Myhre, G., et al. (2016). Recommendations for diagnosing effective radiative forcing from climate models for CMIP6. *Journal of Geophysical Research: Atmospheres*, 121, 12,460–12,475. https://doi. org/10.1002/2016JD025320
- Forster, P. M. D. F., Blackburn, M., Glover, R., & Shine, K. P. (2000). An examination of climate sensitivity for idealised climate change experiments in an intermediate general circulation model. *Climate Dynamics*, 16(10–11).
- Gillett, N. P., Allen, M. R., & Tett, S. F. B. (2000). Modelled and observed variability in atmospheric verical temperature structure. *Climate Dynamics*, 16(1), 49–61.
- Good, P., Gregory, J. M., Lowe, J. A., & Andrews, T. (2013). Abrupt CO₂ experiments as tools for predicting and understanding CMIP5 representative concentration pathway projections. *Climate Dynamics*, 40(3–4), 1041–1053. https://doi.org/10.1007/s00382-012-1410-4
- Gregory, J. M., Andrews, T., Good, P., Mauritsen, T., & Forster, P. M. (2016) Small Global mean cooling due to volcanic radiative forcing. *Climate Dynamics*. 47, 3979–3991. https://doi.org/10.1007/s00382-016-3055-1
 - Gregory, J., & Webb, M. (2008). Tropospheric adjustment induces a cloud component in CO₂ forcing. *Journal of Climate*, 2(1), 1–20.
 - Gregory, J. M. (2004). A new method for diagnosing radiative forcing and climate sensitivity. *Geophysical Research Letters*, *31*, L03205. https://doi.org/10.1029/2003GL018747
 - Hansen, J., Johnson, D., Lacis, A., Lebedeff, S., Lee, P., Rind, D., & Russell, G. (1981). Climate impact of increasing atmospheric carbon dioxide. Science (New York, N.Y.), 213(4511), 957–966. https://doi.org/10.1126/science.213.4511.957



Supercomputing Service. The ECHAM-HAMMOZ model is developed by a consortium composed of ETH Zurich, Max Planck Institut für Meteorologie, Forschungszentrum Jülich, University of Oxford, the Finnish Meteorological Institute, and the Leibniz Institute for Tropospheric Research and managed by the Center for Climate Systems Modeling (C2SM) at ETH Zurich.

- Hansen, J., Sato, M., & Ruedy, R. (1997). Radiative forcing and climate response. Journal of Geophysical Research, 102(D6), 6831–6864. https://doi.org/10.1029/96JD03436
- Hansen, J., Sato, M. K. I., Ruedy, R., Nazarenko, L., Lacis, A., Schmidt, G. A., et al. (2005). Efficacy of climate forcings. Journal of Geophysical Research, 110, D18104. https://doi.org/10.1029/2005JD005776

Haugstad, A. D., Armour, K. C., Battisti, D. S., & Rose, B. E. J. (2017). Relative roles of surface temperature and climate forcing patterns in the inconstancy of radiative feedbacks. *Geophysical Research Letters*, 44, 7455–7463. https://doi.org/10.1002/2017GL074372

Joshi, M. M., Gregory, J. M., Webb, M. J., Sexton, D. M. H., & Johns, T. C. (2008). Mechanisms for the land/sea warming contrast exhibited by simulations of climate change. *Climate Dynamics*, 30(5), 455–465. https://doi.org/10.1007/s00382-007-0306-1

Kamae, Y., & Watanabe, M. (2013). Tropospheric adjustment to increasing CO₂: Its timescale and the role of land-sea contrast. *Climate Dynamics*, 41(11–12), 3007–3024. https://doi.org/10.1007/s00382-012-1555-1

Kasoar, M., Shawki, D., & Voulgarakis, A. (2018). Similar spatial patterns of global climate response to aerosols from different regions. Npj Climate and Atmospheric Science, 1(1), 1–8. https://doi.org/10.1038/s41612-018-0022-z

Knutti, R., Rugenstein, M. A. A., & Hegerl, G. C. (2017). Beyond equilibrium climate sensitivity. Nature Geoscience, 10(10), 727–736. https:// doi.org/10.1038/ngeo3017

- Kummer, J. R., & Dessler, A. E. (2014). The impact of forcing efficacy on the equilibrium climate sensitivity. *Geophysical Research Letters*, 41, 3565–3568. https://doi.org/10.1002/2014GL060046
- Larson, E. J. L., & Portmann, R. W. (2016). A temporal kernel method to compute effective radiative forcing in CMIP5 transient simulations. Journal of Climate, 29(4), 1497–1509. https://doi.org/10.1175/JCLI-D-15-0577.1

Lewis, N., & Curry, J. (2018). The impact of recent forcing and ocean heat uptake data on estimates of climate sensitivity. *Journal of Climate*, 31, 6051–6071. https://doi.org/10.1175/JCLI-D-17-0667.1

Liu, L., Shawki, D., Voulgarakis, A., Kasoar, M., Samset, B. H., Myhre, G., et al. (2018). A PDRMIP multimodel study on the impacts of regional aerosol forcings on global and regional precipitation. *Journal of Climate*, 31(11), 4429–4447. https://doi.org/10.1175/JCLI-D-17-0439.1

Manabe, S., & Stouffer, R. J. (1980). Sensitivity of a global climate model to an increase of CO₂ concentration in the atmosphere. Journal of Geophysical Research, 85(C10), 5529. https://doi.org/10.1029/JC085iC10p05529

Marvel, K., Schmidt, G. a., Miller, R. L., & Nazarenko, L. S. (2015). Implications for climate sensitivity from the response to individual forcings. *Nature Climate Change*, 6(April), 3–6. https://doi.org/10.1038/nclimate2888

Mitchell, J. F. B., Manabe, S., Meleshko, V., Tokioka, T., Baede, A., Berger, A., et al. (1990). Equilibrium climate change and its implications for the future. In J. T. Houghton, G. J. Jenkins, & J. J. Ephraums (Eds.), *Climate change, the IPCC scientific assessment*, (pp. 137–164). Cambridge: Cambridge University Press.

Modak, A., & Bala, G. (2019). Efficacy of black carbon aerosols: The role of shortwave cloud feedback. *Environmental Research Letters*, 14(8), 084029. https://doi.org/10.1088/1748-9326/ab21e7

Modak, A., Bala, G., Caldeira, K., & Cao, L. (2018). Does shortwave absorption by methane influence its effectiveness? *Climate Dynamics*, 51(9-10), 3653–3672. https://doi.org/10.1007/s00382-018-4102-x

- Modak, A., Bala, G., Cao, L., & Caldeira, K. (2016). Why must a solar forcing be larger than a CO₂ forcing to cause the same global mean surface temperature change? *Environmental Research Letters*, *11*(4), 44,013. https://doi.org/10.1088/1748-9326/11/4/044013
- Myhre, G., Forster, P. M., Samset, B. H., Hodnebrog, Ø., Sillmann, J., Aalbergsjø, S. G., et al. (2017). PDRMIP: A Precipitation Driver and Response Model Intercomparison Project—Protocol and preliminary results. *Bulletin of the American Meteorological Society*, 98(6), 1185–1198. https://doi.org/10.1175/BAMS-D-16-0019.1
- Myhre, G., Shindell, D., Bréon, F.-M., Collins, W., Fuglestvedt, J., Huang, J., et al. (2013). Anthropogenic and natural radiative forcing. In Intergovernmental Panel on Climate Change (Ed.), Climate change 2013—The physical science basis, (pp. 659–740). Cambridge: Cambridge University Press. https://doi.org/10.1017/CBO9781107415324.018
- Pithan, F., & Mauritsen, T. (2014). Arctic amplification dominated by temperature feedbacks in contemporary climate models. *Nature Geoscience*, 7(3), 181–184. https://doi.org/10.1038/ngeo2071

Pu, B., & Dickinson, R. E. (2014). Hydrological changes in the climate system from leaf responses to increasing CO₂. Climate Dynamics, 42(7–8), 1905–1923. https://doi.org/10.1007/s00382-013-1781-1

- Ramaswamy, V., Boucher, O., Haigh, J., Hauglustaine, D., Haywood, J., Myhre, G., et al. (2001). Chapter 6: Radiative forcing of climate change. In *Climate change 2001: The scientific basis*, (pp. 349–416). New York: Cambridge University Press.
- Rose, B. E. J., & Rayborn, L. (2016). The effects of ocean heat uptake on transient climate sensitivity. Current Climate Change Reports, 2(4), 190–201. https://doi.org/10.1007/s40641-016-0048-4
- Rotstayn, L. D., Collier, M. A., Jeffrey, S. J., Kidston, J., Syktus, J. I., & Wong, K. K. (2013). Anthropogenic effects on the subtropical jet in the Southern Hemisphere: Aerosols versus long-lived greenhouse gases. *Environmental Research Letters*, 8(1), 14030. https://doi.org/ 10.1088/1748-9326/8/1/014030
- Rotstayn, L. D., Collier, M. A., Shindell, D. T., & Boucher, O. (2015). Why does aerosol forcing control historical global-mean surface temperature change in CMIP5 models? *Journal of Climate*, 28(17), 6608–6625. https://doi.org/10.1175/JCLI-D-14-00712.1
- Rugenstein, M. A. A., Caldeira, K., & Knutti, R. (2016). Dependence of global radiative feedbacks on evolving patterns of surface heat fluxes. *Geophysical Research Letters*, *43*, 9877–9885. https://doi.org/10.1002/2016GL070907
- Santer, B. D., Painter, J. F., Bonfils, C., Mears, C. A., Solomon, S., Wigley, T. M. L., et al. (2013). Human and natural influences on the changing thermal structure of the atmosphere. *Proceedings of the National Academy of Sciences*, 110(43), 17,235–17,240. https://doi.org/ 10.1073/pnas.1305332110
- Santer, B. D., Solomon, S., Pallotta, G., Mears, C., Po-Chedley, S., Fu, Q., et al. (2017). Comparing tropospheric warming in climate models and satellite data. *Journal of Climate*, 30(1), 373–392. https://doi.org/10.1175/JCLI-D-16-0333.1
- Schmidt, H., Alterskjær, K., Bou Karam, D., Boucher, O., Jones, A., Kristjánsson, J. E., et al. (2012). Solar irradiance reduction to counteract radiative forcing from a quadrupling of CO₂: Climate responses simulated by four earth system models. *Earth System Dynamics*, 3(1), 63–78. https://doi.org/10.5194/esd-3-63-2012
- Sherwood, S. C., Bony, S., Boucher, O., Bretherton, C., Forster, P. M., Gregory, J. M., & Stevens, B. (2015). Adjustments in the forcingfeedback framework for understanding climate change. *Bulletin of the American Meteorological Society*, 96(2), 217–228. https://doi.org/ 10.1175/BAMS-D-13-00167.1
- Shindell, D., & Faluvegi, G. (2009). Climate response to regional radiative forcing during the twentieth century. *Nature Geoscience*, 2(4), 294–300. https://doi.org/10.1038/ngeo473

Kang, S. M., & Xie, S. P. (2014). Dependence of climate response on meridional structure of external thermal forcing. *Journal of Climate*, 27(14), 5593–5600. https://doi.org/10.1175/JCLI-D-13-00622.1



- Shindell, D. T. (2014). Inhomogeneous forcing and transient climate sensitivity. Nature Climate Change, 4(4). https://doi.org/10.1038/NCLIMATE2136
- Shindell, D. T., Voulgarakis, A., Faluvegi, G., & Milly, G. (2012). Precipitation response to regional radiative forcing. Atmospheric Chemistry and Physics, 12(15), 6969–6982. https://doi.org/10.5194/acp-12-6969-2012
- Shindell, D. T. D. T., Faluvegi, G., Rotstayn, L., & Milly, G. (2015). Spatial patterns of radiative forcing and surface temperature response. Journal of Geophysical Research: Atmospheres, 120, 1–19. https://doi.org/10.1002/2014JD022752.Received
- Smith, C. J., Forster, P. M., Allen, M., Leach, N., Millar, R. J., Passerello, G. A., & Regayre, L. A. (2018). FAIR v1.3: A simple emissions-based impulse response and carbon cycle model. *Geoscientific Model Development*, 11(6), 2273–2297. https://doi.org/10.5194/gmd-11-2273-2018
- Smith, C. J., Kramer, R. J., Myhre, G., Forster, P. M., Soden, B. J., Andrews, T., et al. (2018). Understanding rapid adjustments to diverse forcing agents. *Geophysical Research Letters*, 45, 12,023–12,031. https://doi.org/10.1029/2018GL079826
- Stjern, C. W., Samset, B. H., Myhre, G., Forster, P. M., Hodnebrog, Ø., Andrews, T., et al. (2017). Rapid adjustments cause weak surface temperature response to increased black carbon concentrations. *Journal of Geophysical Research: Atmospheres*, 122, 11,462–11,481. https://doi.org/10.1002/2017JD027326
- Stuber, N., Ponater, M., & Sausen, R. (2005). Why radiative forcing might fail as a predictor of climate change. *Climate Dynamics*, 24(5), 497–510. https://doi.org/10.1007/s00382-004-0497-7
- Vinnikov, K. Y., Robock, A., Stouffer, R. J., & Manabe, S. (1996). Vertical patterns of free and forced climate variations. *Geophysical Research Letters*, 23(14), 1801–1804. https://doi.org/10.1029/96GL01736
- Wilson, C. A., & Mitchell, J. F. B. (1987). Climate sensitivity experiment with a global climate model including a simple ocean. Journal of Geophysical Research, 92(7), 13,315. https://doi.org/10.1029/JD092iD11p13315
- Winton, M., Takahashi, K., & Held, I. M. (2010). Importance of ocean heat uptake efficacy to transient climate change. *Journal of Climate*, 23(9), 2333–2344. https://doi.org/10.1175/2009JCLI3139.1
- Yoshimori, M., & Broccoli, A. J. (2008). Equilibrium response of an atmosphere-mixed layer ocean model to different radiative forcing agents: Global and zonal mean response. Journal of Climate, 21(17), 4399–4423. https://doi.org/10.1175/2008JCLI2172.1